Author's response to reviews

Title: Challenges and opportunities for implementing evidence-based antenatal care in Mozambique: a qualitative study

Authors:

Adriano Biza (adrianobiza@gmail.com)
Ingeborg Jille-Traas (ingeborg@jille.org)
Mercedes Colomar (mcolomar@unicem-web.org)
Maria Belizan (mbelizan@iecs.org.ar)
Jennifer Harris Requejo (jharris36@jhu.edu)
Beatrice Crahay (icrh.coord@tdm.co.mz)
Mario Merialdi (perialdim@gmail.com)
My Huong Nguyen (nguyenh@who.int)
Fernando Althabe (falthabe@iecs.org.ar)
Alicia Aleman (aaleman@unicem-web.org)
Eduardo Bergel (bergel@gmail.com)
Alicia Carbonell (carbonella@who.int)
Leonardo Chavane (leochavane@gmail.com)
Therese Delvaux (TDelvaux@itg.be)
Diederike Geelhoed (geelhoed.d.w@gmail.com)
Metin Gülmezoglu (gulmezoglu@who.int)
Celsa Regina Malapende (CelsaMalapende@cmam.gov.mz)
Armando Melo (lagicy@yahoo.com.br)
Nafissa Bique Osman (nafissa.osman@gmail.com)
Mariana Widmer (widmerm@who.int)
Marleen Temmerman (temmermanm@who.int)
Ana Pilar Betrán (betrana@who.int)

Version: 4
Date: 2 July 2015

Author's response to reviews: see over
Dear Editor,

Many thanks for your response.

As requested we have included details of the source of funding in the Acknowledgement section. We have also revised and formatted the manuscript to conform with the journal style.

Kind regards,

Ana Pilar Betrán

Ana Pilar Betrán
Maternal and Perinatal Health and Preventing Unsafe Abortion
HRP – UNDP · UNFPA · UNICEF · WHO · World Bank Special Programme of Research, Development and Research Training in Human Reproduction
Department of Reproductive Health and Research
World Health Organization, Avenue Appia 20, Geneva 27
CH-1211 Switzerland
Tel: +41-22-7914205
Fax: +41-22-7914171
E-mail: betrana@who.int